Author's response to reviews

Title: The authors reply to the paper entitled Structure, (governance) and health: an unsolicited response by Daniel D Reidpath and Pascale Allotey

Authors:

Anatole S Menon-Johansson (anatole_uk@yahoo.co.uk)

Version: 2 Date: 26 July 2006

Author's response to reviews:

Dear Emma,

Please could you remove the abstract and place the figure in the appropriate place.

I do not mind if you would prefer to omit the title.

Many thanks,

Anatole